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Operations | - summary >
" N
J BEAM:
e Beam for NOvA except one day shutdown lastTue and today
morning.
J NearDet:
° During shutdown many APD/FEBs were swapped to keep noise
lower.
JFarDet:

° Physics with 14-diblocks running since Nov 20.

° During the maintenance day Ash River check one Dry gas
system and due to swap 20 APDs and other 8 through detector.

e Otherwise the detector was running stable.

J Control Room:

v" Satisfied with new Remote control room in Wichita.
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FarDet: Data-Taking s

Hit rate recorded by FEB in Hz
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Partition 1 (status since Nov 20)
Current status — Retrofitting: DONE!
= diblocks 1 — 14

" Running ~99% active,

= ALL COOLED, HV at full gain. go bellow alarm)
Jaroslav Zalesak - NOvA Collaboration Meeting 10/24/2014

Maintenance day: checked and
repaired Dry Gas system 6V2, dcm-2-
06-{03,04} (low pressure - potential

automatically power shut-off when
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FD: Data taking Uptime/POTs A

Fa “ASRL VLN

We have trouble with our Run Info database. It looks like we are missing information about runs
since Thursday last week.
We are workin g on it. Nevertheless our best expert is travelin g for a Conference in Mexico and

internet connection is bad.

»In general we had another good week, neglecting the scheduled beam shutdown.

» Mon: we had 100% uptime.

» Tue: we powered down (partially) detector when beam shut down and started
maintenance, resulted in ~13 hours downtime, nevertheless NO beam at all at that times.
» Wed: we needed to update new HYV settings for channels where the electronics were
replaced the day before (safety rules) and take new DSO (different working position).
At that point we lost about 2.5hours of uptime and -0.15e¢18 POTs

» Rest of week almost 100% efficient! Beam lost only 24 min! Estimated average value is -
0.022¢18 POTs
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